Author's response to reviews

Title: Recurrent White Thrombi Formation in Hemodialysis Tubing Lines: A Case Report

Authors:

Kiran P Sathe (kiranpsathe@yahoo.co.in)
Wee-Song Yeo (cornelius_lionel@yahoo.com)
Isaac Desheng Liu (mynjournalclub@yahoo.com)
Sudha Ekambaram (docsudha80@yahoo.co.in)
Mohammed Azar (shameemazar@gmail.com)
Hui-Kim Yap (hui_kim_yap@nuhs.edu.sg)
Kar-Hui Ng (paenkh@nus.edu.sg)

Version: 2
Date: 17 November 2014

Author's response to reviews:

Hi,

As commented in the email dated 15th Nov 2014, formatting edits such as inserting line numbers and removing the page breaks done as on 17 Nov 2014. The updated manuscript uploaded again on the portal. The rest all detail including the cover letter remain as in original submission.

Thank you.